Author's response to reviews

Title: Serum Thyroid-Stimulating Hormone Levels are not Associated with Exercise Capacity and Lung Function Parameters in two Population-Based Studies

Authors:

Till Ittermann (till.itterman@uni-greifswald.de)
Sven Gläser (sven.glaeser@uni-greifswald.de)
Ralf Ewert (ewert@uni-greifswald.de)
Stephan Felix (felix@uni-greifswald.de)
Henry Völzke (voelzke@uni-greifswald.de)
Marcus Dörr (mdoerr@uni-greifswald.de)

Version: 2 Date: 3 February 2014

Author's response to reviews: see over
Dear Prof. Zapanta,

we now provide a Conclusion after the Discussion and renamed the Competing Interests section.

Yours sincerely

Till Ittermann